Author's response to reviews

Title: Do health technology assessments comply with QUOROM diagram guidance? : an empirical study

Authors:

Daniel Hind (d.hind@sheffield.ac.uk)
Andrew Booth (a.booth@sheffield.ac.uk)

Version: 4 Date: 30 October 2007

Author's response to reviews: see over
Dear Editor

Thank you for agreeing to publish our article.

David Moher made two suggestions and we have acted on both.

We have rearranged the box contents in line with David’s request.

We have replaced the word ‘moderate’ with the word ‘poor’

Yours sincerely,

Daniel Hind
Research Fellow